Author’s response to reviews

Title: Diet satisfaction and associated factors among adult surgical orthopaedic patients admitted at a teaching hospital in Lusaka province, Zambia; A hospital-based cross-sectional study

Authors:

Nixon Miyoba (nixonmiyoba@gmail.com)
Irene Ogada (irene.a.ogada@gmail.com)

Version: 6 Date: 22 Feb 2019

Author’s response to reviews:

Kitwe Teaching Hospital,
P.O. Box 20969.
Kitwe.
22/02/2019
The Editor,
BMC Journal
Dear Sir/Madam,
Re: NUTN-D-17-00206R5

Comment No.1: Please separate the Availability of data and materials subsection from the Consent for publication subsection. Response No.1: The separation has been effected.

Comment No.2: Submission of a manuscript to a BIOMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them. Response No. 2: The questionnaire has been uploaded.

Comment No. 3: Please remove all information related to funding from the Acknowledgments subsection. Funding information should be presented solely in the Funding subsection. The Acknowledgments subsection should solely feature information which acknowledges anyone who contributed towards the article who does not meet the criteria for authorship including
anyone who provided professional writing services or materials. Response No. 3: The information related to funding has been removed as suggested.

Nixon Miyoba